

October 10, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 12491-review.doc).

**Title:** Recent Advances in Management of Retinoblastoma-A Review

**Authors:** Bhavna Chawla, Gautam Lokdarshi, Sushmita Pathy

**Name of Journal:** *World Journal of Ophthalmology*

**ESPS Manuscript NO:** 12491

The manuscript has been revised in accordance with Reviewer's and Editor's comments.

1. Page 6, Section iii on Sub-conjunctival/Sub-tenon Chemotherapy: A definition of Group C and D tumors has been included

2. Format has been updated

3. The following information has been added-

- (1) Post-codes of all authors have been mentioned
- (2) Telephone and fax numbers of corresponding author have been added
- (3) Core tip/summary of less than 100 words has been added
- (4) Reference numbers have been added in square brackets as superscript

4. PMID, DOI and all authors have been mentioned in the references. Typesetting has been done

5. Grammatical errors have been corrected

Thanking you for considering our work for publication in the *World Journal of Ophthalmology*,

With best regards,

Dr. Bhavna Chawla

Additional Professor

Ocular Oncology & Pediatric Ophthalmology Service

Rajendra Prasad Centre for Ophthalmic Sciences

All India Institute of Medical Sciences

New Delhi, India

Email: bhavna [2424@hotmail.com](mailto:bhavna_2424@hotmail.com)

Telephone: +91-11-26593088, Fax: +91-11-26588919